Author's response to reviews

Title: Factors influencing the effectiveness of multisource feedback in improving the professional practice of medical doctors: A systematic review

Authors:

Julie Ferguson (julie.ferguson@nes.scot.nhs.uk)
Judy Wakeling (judy.wakeling@nes.scot.nhs.uk)
Paul Bowie (paul.bowie@nes.scot.nhs.uk)

Version: 3 Date: 28 March 2014

Author's response to reviews:

Authors response to reviews

The authors would like to thank the editors for their review of the manuscript MS: 6151596911578223 - Factors influencing the effectiveness of multisource feedback in improving the professional practice of medical practitioners: A systematic review. We have made the final change regarding in accordance with the editorial request regarding ensuring that the manuscript reporting adheres to PRISMA guidelines and completion of the PRISMA checklist. The checklist has now been completed - however due to the nature of the studies (only one RCT) included in the review some of the checklist items are not relevant, particularly 13, 19, 20 and 21. We have also added a statement on page 5 "This study was conducted and reported according to PRISMA guidelines".